REVIEW

Preferences related to attention-deficit/ hyperactivity disorder and its treatment

Kate Van Brunt¹ Louis S Matza¹ Peter M Classi² Joseph A Johnston²

¹Center for Health Outcomes Research at United BioSource Corporation, Bethesda, MD, USA; ²Eli Lilly and Company, Indianapolis, IN, USA

Correspondence: Louis S Matza Center for Health Outcomes Research at United BioSource Corporation, 7101 Wisconsin Avenue, Suite 600, Bethesda, MD 20814, USA Tel +1 301 664 7263 Fax +1 301 654 9864 Email louis.matza@unitedbiosource.com

submit your manuscript | www.dovepress.com Dovepress DOI: 10.2147/PPA.S6389 **Objectives:** A growing body of literature has highlighted the importance of considering patient preferences as part of the medical decision-making process. The purpose of the current review was to identify and summarize published research on preferences related to attention-deficit/hyperactivity disorder (ADHD) and its treatment, while suggesting directions for future research.

Methods: A literature search identified 15 articles that included a choice-based assessment of preferences related to ADHD.

Results: The 15 studies were grouped into four categories based on preference content: preference for a treatment directly experienced by the respondent or the respondent's child; preference for general treatment approaches; preference for a specific treatment attribute or outcome; and preference for aspects of ADHD-related treatment. Preference assessment methods ranged from global single items to detailed choice-based procedures, with few studies using rigorously developed assessment methods. Respondents included patients with ADHD, clinicians, parents, teachers, and survey respondents from the general population. Factors influencing preference include treatment characteristics, effectiveness for specific symptoms, side effects, and respondent demographics. Minimal research has examined treatment preferences of adults with ADHD.

Discussion: Because there is no dominant treatment known to be the first choice for all patients, ADHD is a condition for which individual preferences can play an important role when making treatment decisions for individual patients. Given the potential role of preferences in clinical decision-making, more research is needed to better understand the preferences of patients with ADHD and other individuals who are directly affected by the disorder, such as parents and teachers.

Keywords: patient preference, ADHD, parent preference, utility

Objectives

A growing body of literature has highlighted the importance of considering patient preferences as part of the medical decision-making process.^{1–3} Consequently, studies have assessed patient preferences for treatment options across a wide range of medical and psychiatric conditions such as cancer,^{4–6} allergic rhinitis,⁷ depression,⁸ migraine,⁹ diabetes,¹⁰ and osteoporosis.¹¹ Several studies have examined preferences for treatment of attention-deficit/hyperactivity disorder (ADHD), which is characterized by symptoms of inattention, hyperactivity, and impulsivity.¹² Research on preferences for treatment of childhood ADHD raises methodological questions as studies often examine preferences of individuals other than the children themselves, such as parents, teachers, clinicians, and the general public. In addition, although awareness of adult

Patient Preference and Adherence 2011:5 33-43

© 2011 Van Brunt et al, publisher and licensee Dove Medical Press Ltd. This is an Open Access article which permits unrestricted noncommercial use, provided the original work is properly cited.

ADHD is growing,^{13–16} little is known about preferences of adults with ADHD. Thus, the purpose of the current review was to identify and summarize published research on preferences related to ADHD and its treatment, while suggesting directions for future research.

Background: patient preference

Despite the range of definitions in the literature, there appears to be a consensus that the term 'patient preference' refers to a patient's perception of the relative desirability of more than one health-related option.^{17–20} Research has been conducted to identify and quantify patient preferences for a wide range of treatment options and health states. For example, some studies have asked patients to indicate which treatment option they preferred after receiving multiple treatments in a clinical trial with a crossover design.^{7,9,21} Research participants have also been asked to express preferences among hypothetical health states that they have not necessarily experienced.²² Preferences of individual patients may also be considered in clinical settings as part of a shared patient–clinician decisionmaking process.¹⁷

Patient preference is considered important for several reasons. First, there is growing awareness that active patient participation in the medical decision-making process may have potential treatment benefits.^{17,18} Patients often want to be involved in these decisions,²³ and studies have found that greater patient involvement in health care decisions may be associated with increased treatment adherence, symptom relief, and treatment satisfaction.8,24-27 When making treatment decisions with individual patients, the patient's preferences are likely to be consistent with evidence-based medicine and generally accepted clinical practices.³ However, there are circumstances when individual patient preferences diverge from those of health professionals and the general public.19 Individual patient preferences may be most important when clinical trial results have not yet indicated which treatment option tends to be more effective or when similarly effective treatment options could have different effects on quality of life.17

Patient preference data collected in clinical trials and other studies could substantially contribute to large-scale health care decision-making. Patient preferences identified within studies involving larger samples can provide an indication of comparative treatment effectiveness in the total sample and among meaningful patient subgroups, which may help guide clinicians when deciding how to treat individual patients. In addition, preference data may shape broader treatment recommendations, as these data provide an indication of the patient's perspective that could be used by decision-makers when drafting treatment guidelines and health policy.¹⁷ Patient preferences are also used to quantify the health-related quality of life of health states. These resulting estimates called utilities, with values of 1 corresponding to full health and 0 corresponding to death, quantify health outcomes and treatment benefits and are used in cost-utility analyses which inform medical decision-making.^{28,29}

Patient preferences are assessed with a wide variety of methods, ranging from global items to more detailed choicebased assessment methods. Some studies, including many clinical trials, have used straightforward single items asking patients which they prefer among two or more treatment options. In clinical trials with crossover designs, these single items may be used to assess preferences among treatments that the patients have recently experienced.³⁰⁻³² In other studies, patients may be asked to express preferences among a range of treatments or health-related options that they have not personally experienced.^{33–36} Occasionally, global preference questions may be followed by Likert scale items assessing the strength of preference for the various options.¹⁹ Studies aiming to quantify preferences in terms of utilities often use more complex methodology involving choices between hypothetical health state options. These methods, such as standard gamble (SG) and time-trade-off procedures, have been summarized previously.^{29,37,38} The current review summarizes literature using any of these methods to assess preferences associated with ADHD.

Literature review methods

A literature search was conducted using the PubMed database with no restrictions on date of publication. An initial search for citations mentioning ADHD or any terms related to the disorder yielded 13,495 citations. Then, a second search identified articles relating to preference, using search terms corresponding to preference in a general sense, terms referring to a specific preference assessment method, and terms that could be related to preference such as acceptability and decision-making. These search terms included all forms (eg, singular and plural, abbreviations, alternative spellings, noun, and verb forms) of the following: health state utility, utility, discrete choice, standard gamble, time trade-off, quality-adjusted life year, conjoint analysis, patient preference, preference, prefer, satisfaction, acceptability, decision, and choice. The preference search, which yielded 757,804 citations, was then crossed with the ADHD search resulting in 1005 abstracts.

The 1005 abstracts, and full-text articles when necessary, were reviewed to identify articles meeting inclusion/exclusion criteria. For this literature review, 'preference' was conceptualized based on the definition proposed by Brennan and Strombom:18 'statements made by individuals regarding the relative desirability of a range of health experiences, treatment options, or health states'. For a study to be considered a 'preference study', it was required that respondents were given a choice among multiple health-related options. Questionnaires or interviews assessing perceptions of a single treatment without comparison to an alternative option were not considered to be preference assessments, even if articles used terms that initially appeared to be relevant, such as 'prefer' or 'choice'. Both informal methods (eg, unvalidated single items or interviews) and formal methods (eg, SG, time-trade-off, and discrete choice experiments) for assessing preference were included. Preferences of children with ADHD, adults with ADHD, parents, teachers, clinicians, and the general public were all considered to be relevant for the current review.

The following citations were excluded: review articles, conference presentations, letters, practice guidelines, case studies, and editorials. Cost-effectiveness and cost-utility analyses were not included, but these articles were examined in order to identify any utility or preference data that may have been cited. Articles focusing on conceptually related topics such as treatment acceptability, treatment satisfaction, treatment-related attitudes, and decision-making were excluded if respondents were not asked to indicate a preference among multiple health-related options. Although multiattribute measures such as the EQ-5D® and Health Utilities Index have scoring algorithms that were derived via preference-based tasks, studies administering these instruments were not included in the current review because respondents do not explicitly indicate preferences when completing these questionnaires.

Results: ADHD preference studies Summary of preference studies

A total of 15 articles were identified that included a choicebased assessment of preference related to ADHD. For the current review, these 15 studies are organized into four categories based on the content of the preference assessment: 1) five studies assessing preference for a treatment directly experienced by the respondent or the respondent's child with ADHD, 2) four studies assessing preference for general treatment approaches, 3) four studies assessing preference for a specific treatment attribute or outcome, and 4) two studies that did not fit into the three other categories because they did not examine preference for a treatment-related aspect of ADHD.

Seven studies that were excluded from this review used the term 'preference' when describing methods or results, but did not appear to include a choice-based assessment of preference among health-related options. Three of these seven studies used qualitative interview or focus group methods to elicit open-ended responses,^{39–41} and the remaining four studies administered rating scales that assessed related constructs such as importance and acceptability.^{42–45} These seven studies were excluded from the current review because they were not consistent with generally accepted definitions of 'preference', which involves a choice between two or more options.

Studies assessing preference for a specific experienced treatment

Five studies assessed preference for a treatment directly experienced by the respondent or the respondent's child (Table 1). Three of the five studies presented results from clinical trials of medication treatment for ADHD in children and/or adolescents.46-48 Because these studies focus on efficacy and safety of medication treatment, the published articles do not provide a detailed description of the preference assessment methods. The study by Efron et al⁴⁶ specified that a single-item assessment was completed by parents of children treated for ADHD, while the articles by Quintana et al⁴⁷ and Pelham et al⁴⁸ do not specify the preference assessment method. In the study by Efron et al,⁴⁶ the single-item assessment was completed at the end of the 4-week crossover trial of methylphenidate (MPH) and dexamphetamine (DEX), with findings indicating that more parents preferred the 2-week MPH treatment period over the 2-week DEX treatment period (46.6% vs 36.8% of parents). In the placebo-controlled, three-way crossover trial by Pelham et al,48 children received treatment with immediaterelease (IR) MPH three times daily, MPH once daily, and placebo, each for a 7-day period. Results from the unspecified preference assessment completed by parents at the end of the three treatment periods found that 47% of parents selected the once daily MPH formulation as the treatment of choice for their child versus 31% of parents who selected the immediate-release formulation taken three times daily. Finally, the study by Quintana et al⁴⁷ presents results from a 6-week clinical trial in which children and adolescents switched from psychostimulant treatment to treatment with atomoxetine. Although the preference assessment method and preference evaluator were not clearly specified in the

Table I Studies assessing preference for treatments directly experienced by the respondents or their children

Citation	Preference assessment method	Preference content	Respondent (N)	Key results
Efron et al ⁴⁶	Single-item assessment of preference at end of treatment period in a crossover trial (no further details provided)	Children received treatment with MPH and DEX, each for 2 weeks in a double-blind, crossover trial. After both treatment periods, parents were asked to specify which treatment they preferred	Parents of children (mean age of 8.73 years) with ADHD (N = 125)	46 of 104 parents (36.8%) indicated that they preferred the DEX treatment period, and 58 parents (46.4%) indicated that they preferred the MPH treatment period
Fredericks and Kollins ⁴⁹	Double-blind choice procedure	Participants received double-blind treatment with either placebo or MPH during four study 'sampling sessions'. During eight subsequent 'choice sessions', participants chose which treatment they would receive: placebo, MPH, or neither	Adults with ADHD (N = 10)	MPH was preferred in 50% of the choices; placebo 32.5%, neither 17.5% (significant difference among choices; $\chi^2 = 52.5$, $P < 0.001$)
MacDonald Fredericks and Kollins ⁵⁰	Double-blind choice procedure	Participants received double-blind treatment with either placebo or MPH during six study 'sampling sessions'. During six subsequent 'choice sessions', participants chose which treatment they would receive: placebo, MPH, or neither	Children/adolescents (aged 10–14) with ADHD (N = 5)	MPH was preferred in 60% of the choices; placebo 20%; neither 20% (significant difference among choices; $\chi^2 = 9.6, P < 0.01$)
Pelham et al ⁴⁸	Unspecified preference assessment completed at end of double-blind, placebo-controlled, clinical trial	Children received treatment with placebo, IR MPH (three times daily), and a once daily MPH formulation in a randomly selected order. At the end of the three 7-day treatment periods, parents were asked to choose which of the treatment weeks they preferred for their child	Parents of children with ADHD (aged 6–12) (N = 68)	47% of the parents selected once daily MPH as the treatment of choice, 31% selected IR MPH, 15% chose their child's previous MPH treatment, and 7% either chose placebo or had no preference
Quintana et al ⁴⁷	Unspecified preference assessment completed at end of clinical trial	In this 6-week study, children and adolescents with incomplete response or intolerance to stimulant treatment switched to atomoxetine after the first week. The two treatments were compared using an unspecified preference assessment method	Children and adolescents (aged 6–17 years) with ADHD (N = 58)	65.5% of subjects reported a preference for atomoxetine treatment over their previous psychostimulant

Abbreviations: MPH, methylphenidate; DEX, dexamphetamine; ADHD, attention-deficit/hyperactivity disorder; IR, immediate release.

study results, the abstract of this article reported that 65.5% of respondents expressed a preference for atomoxetine treatment over their psychostimulant.

The remaining two studies assessing preference for a directly experienced treatment were double-blind choice procedures performed in small samples of adults (N = 10)⁴⁹ and children/adolescents (N = 5)⁵⁰ who were receiving treatment with MPH at the time of enrollment in the study. In both studies, participants received double-blind treatment with either placebo or MPH during each of the study 'sampling sessions', which were followed by the 'choice sessions' in

which participants were asked to choose which treatment they would receive. Treatment options at these sessions included placebo, MPH, or neither treatment, with authors considering each participant's choice to be an indicator of drug preference. In the choice procedure conducted with adult patients, MPH was chosen as treatment 50% of the time, placebo was chosen 32.5% of the time, and neither treatment was chosen 17.5% of the time, with the difference among treatment choices being significant ($\chi^2 = 52.5$, P < 0.001). In the study conducted with children, differences were also significant among treatment choices, with participants choosing MPH 60% of the

time, placebo 20% of the time, and neither treatment 20% of the time ($\chi^2 = 9.6, P < 0.01$).

Although heterogeneity in study designs, variation in preference assessment methods, and differences in the ADHD treatments make it difficult to draw overall conclusions from these five studies, some general trends did emerge. Results of the double-blind choice procedures suggest a preference for MPH over placebo among adults and children who received both treatments, while the crossover trial by Efron et al⁴⁶ found that parents prefer MPH over DEX as treatment for their children. Results from the double-blind trial by Pelham et al⁴⁸ suggest that less frequent dosing may be preferable among parents, while the study by Quintana et al⁴⁷ suggests that a nonstimulant treatment might be preferable for some children and adolescents. Overall, these studies indicate that a preference assessment may be a useful approach for quantifying and comparing patients' or parents' experiences with drug treatments.

Studies assessing preference for general treatment approaches

Table 2 presents results from the four studies assessing preference for a general treatment approach.^{51–54} All four studies assessed preference using a survey or questionnaire. In three of the studies, participants responded by indicating their choice among multiple options. In the study by McLeod et al,⁵³ participants were asked yes/no questions regarding their opinions of counseling and medication treatment for ADHD. The authors then derived preferences based on the pattern of responses to these two questions. These four studies were conducted in samples of parents, teachers, and the general public. Across the four studies, the treatment approaches under investigation included medication-only regimens, nonmedication regimens (eg, counseling or behavior modification approaches), and combined approaches of medication and nonmedication treatments.

Results of these four studies generally suggested that combined treatment approaches may be preferred to monotherapy treatment approaches for children with ADHD, but there is some variability in preferences. Three of the four studies found that a majority of respondents chose a combined treatment approach (ie, medication plus counseling or behavior modification) over a monotherapy treatment approach.^{52–54} The respondents varied across these three studies, with samples consisting of teachers, the general public, and ethnically diverse parents of children with and without ADHD. The questionnaire included in the fourth study by Dos Reis et al⁵¹ included an item relating to preference, with authors reporting results specific to racial-ethnic comparisons groups. Findings suggested that nonwhite parents were less likely than white parents to prefer medication over counseling as a treatment option for children with ADHD (59% of white parents vs 36% of nonwhite parents, P < 0.0001).

Studies assessing preference for treatment attributes or treatment outcomes

Four studies were identified that assessed preference for treatment attributes or outcomes (Table 3). Unlike the studies presented in Table 1, participants in these studies were not asked to report preferences for treatments that they, or their children, directly experienced. Instead, respondents were asked to indicate a preference for attributes or outcomes relating to hypothetical treatment choices. In these four studies, preference was assessed by a discrete choice experiment,⁵⁵ a SG utility assessment interview,^{56,57} or a survey mail-out.⁵⁸ Across the three studies involving a formal preference procedure (ie, discrete choice or SG), samples included parents of children or adolescents with ADHD. The sample in the study by Stockl et al⁵⁸ consisted of 365 physicians who were treating children and adolescents with ADHD.

Several treatment attributes and outcomes were assessed in these studies, including the type of treatment (eg, stimulant vs nonstimulant), duration of effect, side effect profile, overall treatment efficacy, and impact on school and family functioning. Although the attributes and outcomes varied across these four studies, there was some consistency in results. Results from three studies suggested that nonstimulants may be preferred over stimulants for the treatment of children with ADHD.56-58 In the utility studies by Matza et al⁵⁶ and Secnik et al,⁵⁷ parents expressed their preference for a nonstimulant treatment option over a stimulant treatment option when both hypothetical treatments were otherwise equal in terms of efficacy, side effect profile, and other treatment attributes. The survey results reported in Stockl et al⁵⁸ found that 38% of physicians strongly agreed or agreed that they would prefer prescribing a nonstimulant instead of a stimulant for the treatment of ADHD in children, provided that such options are available and Food and Drug Administration-approved. However, because the respondents did not necessarily have direct experience with nonstimulant medications and they were not provided with information on risks and benefits of stimulant treatment, these findings likely represent preconceived biases rather than preferences based on direct experience.

Table 2 Studie	s assessing pre	ference for gene	eral treatment approa	ches
----------------	-----------------	------------------	-----------------------	------

Citation	Preference assessment method	Preference content	Respondent (N)	Key results
Dos Reis et al ^{s1}	ASK-ME survey (a 47-item, self-administered questionnaire)	Parents recruited from six pediatric primary care clinics completed the ASK-ME. One item asks respondents to indicate level of agreement with the following statement: '[1] prefer medication over counseling'. Study results were presented by racial/ethnic comparison groups (white vs nonwhite parents)	Parents of youth diagnosed with ADHD (N = 254)	Nonwhite parents were less likely than white parents to 'prefer medication over counseling' for their children (59% of white parents vs 36% of nonwhite parents, P < 0.0001)
Glass and Wegar ⁵²	Surveys distributed to teachers	Surveys assessed teachers' perceptions of ADHD etiology and treatment options. Teachers were given a choice of the following treatment options: medication, behavior modification, medication plus behavior modification, and no treatment	Teachers of children in kindergarten through fifth grade (N = 225)	94.7% of respondents (N = 213) chose the 'medication and behavior modification' option as the most appropriate treatment regimen
McLeod et al ⁵³	A short battery of questions included in the 2002 General Social Survey, followed by face-to-face interviews	A subset of respondents to the 2002 General Social Survey's National Stigma Study (Children module) who had indicated a prior knowledge and awareness of ADHD and participated in follow-up face-to-face interviews. Interviews included yes/no questions relating to ADHD beliefs and treatment preferences, including 'Should children be given counseling for ADHD?' and 'Should children be given medication to treat ADHD?' Authors used respondents' answers to the above yes/no questions to indicate beliefs and treatment preferences	General public survey respondents (N = 725)	Most respondents believed that children with ADHD should be given a combination of counseling and medication (65%, N = 471). 21% expressed a preference for counseling only (N = 151), 5% expressed a preference for medication only (N = 39), and 9% (N = 64) indicated that children with ADHD should receive neither counseling nor medication
Pham et al ⁵⁴	Questionnaire developed specifically for this study	Parents of children with and without ADHD completed a survey on ADHD- related beliefs and treatment. The survey included an item on treatment preference that asked parents to select from one of the following treatment options: medication only, counseling only, and a combined treatment approach. If parents did not have a child with ADHD, they were provided with a hypothetical situation in which their child did have ADHD	Ethnically diverse parents of children (aged 5–12) with ADHD (N = 58) and without ADHD (N = 61)	53.8% of parents preferred a combined treatment approach for their child, 24.4% preferred counseling only, 16.8% preferred medication only, and 5.0% responded 'none of the above' to the provided treatment options

Abbreviations: ASK-ME, attitudes, satisfaction, knowledge, and medication experiences survey; ADHD, attention-deficit/hyperactivity disorder.

Two additional trends that emerged across these studies were a preference for treatments with no known abuse potential and a preference for treatments with better (ie, more tolerable) side effect profiles. Physicians completing the survey administered by Stockl et al⁵⁸ and parents participating in the discrete choice experiment described by Muhlbacher et al⁵⁵ indicated their preference for treatments with no known abuse potential over treatments with evidence of abuse potential. Parents participating in the utility study by Matza et al⁵⁶ and the discrete choice experiment by Muhlbacher et al⁵⁵ indicated that the side effect profile of a hypothetical ADHD medication was important in the selection of and preference for an ADHD treatment. Specific side effects that influenced preference in these studies included incidence of nausea, changes in weight and appetite, and whether the medication made the children feel drowsy or more 'wired'.

Apart from the trends that emerged in the treatment attributes discussed above, the discrete choice experiment

Citation	Preference assessment method	Preference content	Respondent (N)	Key results
Matza et al ⁵⁶	SG utility interviews	Parents of children with ADHD indicated preferences among 11 hypothetical health states that contained variations of the following five domains: ADHD symptom profile (mild, moderate, and severe symptoms), typical ADHD behaviors, impact on school functioning, impact on family functioning, and type of treatment with corresponding side effect profiles (stimulant, nonstimulant, and no treatment)	Parents of children (mean age of 10.2 years) diagnosed with ADHD (N = 43)	When both treatments were associated with an adequate response and tolerable side effects, parents significantly preferred the nonstimulant health state over the stimulant health state ($t = 2.3$, $P = 0.03$). When either the treatment response was inadequate or the side effects were intolerable, there were no significant differences in preferences for the stimulant and nonstimulant health states
Muhlbacher et al ^{ss}	DCE and questionnaire	Parents' preferences of therapy characteristics were evaluated using a DCE and a 23-item questionnaire assessing importance of ADHD-therapy characteristics (including duration of effect, impact on school performance, and dosing options). The following therapy characteristics were included in the DCE: duration of treatment effect (long vs short), side effects (weight loss vs none), dosage form (always the same vs variable/combinable), discretion (intake of drug obvious vs none), social situation (problems with friends, hobbies vs no problems)	Mothers and fathers of adolescents (mean age 15 years) with ADHD (N = 219)	Results from the 23-item questionnaire found the following therapy characteristics to have the greatest relevance to parents: 'improving the child's emotional state', 'little or no addictive potential', and 'improved ability to concentrate'. Results of the DCE found the following characteristics to influence the selection of treatment: 'enabling social contacts' and 'emotional state: no mood swings' (relative importance 40%). 'Duration of effect: long (all day)' was also desirable (relative importance 18%), as were the characteristics of 'discretion', 'dosage', and 'side effects'
Secnik et al ⁵⁷	SG utility Interviews	Parents indicated preferences among 14 hypothetical health states,which described ADHD-related characteristics that varied according to treatment option (untreated ADHD, ADHD treated with a nonstimulant, immediate-release stimulant, or extended-release stimulant) and the nature of response (responder or nonresponder, with or without side effects)	Parents of children (mean age of 12.6 years) with ADHD (N = 83)	Generally, nonstimulant health states were preferred over otherwise identical stimulant health states
Stockl et al ⁵⁸	Survey mailing of a 30-item questionnaire	A survey was mailed to 1000 physicians who had prescribed stimulant medications to children and adolescents. The following preference-related questions asked clinicians to indicate their level of agreement with the following statements: Question 17: 'If available and with a FDA indication for treating ADHD in children or adolescents, I would prefer prescribing a medication that is not a stimulant versus a stimulant'. Question 18: 'If available and with a FDA indication for treating ADHD in children or adolescents, I would prefer prescribing a noncontrolled medication that does not have evidence of abuse potential versus one that is controlled [and has]	Physicians treating children and adolescents with ADHD (N = 365)	38% of physicians strongly agreed or agreed that they would prefer prescribing a nonstimulant instead of a stimulant if a nonstimulant with an FDA indication were available. 58% of physicians strongly agreed or agreed that they would prefer prescribing a noncontrolled medication that does not have evidence of abuse potential versus one that is controlled and has evidence of abuse potential

by Muhlbacher et al⁵⁵ assessed additional treatment characteristics that were not investigated in the other studies. This study found that treatments with a longer duration of action, greater potential for improvements in emotional state, and enhanced ability to enable social contacts would have the most influence in the parents' selection of treatment for their children with ADHD. These findings suggest that real-world outcomes, in addition to treatment efficacy, contribute to preferences for their children's treatment.

Studies not assessing a treatment-related aspect of ADHD

Finally, two additional studies were located that assessed preference for an aspect of ADHD that was not related to treatment. One study recruited a sample of 99 parents of children with ADHD to complete a survey assessing the importance placed on types of ADHD information and the preferred modes of receiving this information.⁵⁹ Parents were asked to preferentially rank the following ways of receiving information about their child's ADHD: verbal, written, DVD/video, seminars, parenting class, audio, Internet, video, and CD-ROM. Authors found the most preferred mode of information delivery to be verbal information received directly from a professional, with written information being the second most preferred option.

Another study involved semistructured follow-up interviews with 19 teachers of elementary school students with ADHD who had participated in a 2-month clinical trial of an unspecified ADHD treatment.⁶⁰ The teachers were asked to compare the Web-based ADHD symptom rating scale that they completed during the trial (the T-SKAMP) to their previous experience with paper-and-pencil ADHD scales. Results of the interviews found teachers to generally prefer the Web-based scale, with 89.5% of teachers indicating that it was easier to complete than the paper-based scale.

Discussion

40

ADHD may be treated with a range of potentially effective pharmacological and behavioral treatment options. Because there is not a dominant treatment known to be the first choice for all patients, ADHD is a condition for which individual preferences can play an important role when determining a treatment approach for individual patients. In studies identified for the current review, a wide range of measurable treatment preferences were reported by patients with ADHD, clinicians, parents, teachers, and survey respondents from the general population. However, this literature search found only 15 studies using a choice-based preference assessment related to ADHD. Given the potential role of preferences in clinical decision-making, more research is needed to better understand the preferences of patients with ADHD and other individuals who are directly affected by the disorder, such as parents and teachers.

Five studies were identified that assessed preference between two treatment options directly experienced by the respondent or the respondent's child, and all five studies yielded clear preferences (Table 1). Parents expressed preferences among stimulant treatment options,^{46,48} children expressed preferences for a nonstimulant over a stimulant,47 and small samples of children and adults expressed preferences for MPH over placebo.49,50 One limitation of the current review is that clinical trials assessing preferences among ADHD treatment options would not have been located if they did not mention 'preference' or a related term in the published abstract. Therefore, it is possible that the current literature search failed to identify some published clinical trials that included a preference measure, but did not mention it in the abstract. Despite this limitation, results of the five identified studies suggest that preference data can complement clinical symptom measures by providing insight into the experiences of individuals directly affected by treatments. Based on these five studies, assessment of preference can be recommended for inclusion as an outcome measure in future clinical trials with study designs that allow patients to experience more than one treatment option. These preference assessments are more likely to yield useful results if the assessment tools are carefully developed and validated in the target population.

Additional studies assessed preferences for treatment approaches and attributes among respondents who did not necessarily have recent direct experience with the treatment options. Although these preferences were not assessed in the context of a controlled clinical trial, results may still provide useful information for clinical decision-makers. For example, parents, teachers, and general public survey respondents expressed preferences for combined treatment approaches involving both medication and nonpharmacological treatment such as counseling and behavioral modification.⁵²⁻⁵⁴ Three additional studies revealed preferences for nonstimulant medications over stimulants among clinicians and parents.56-58 Another study identified several therapy characteristics that may influence parents' treatment preferences, such as addictive potential, improvement in concentration, effects on social functioning, emotional impact, duration of effect, dosage, and side effects.⁵⁵ Finally, one study found that parent preferences for medication and counseling may vary as a function of racial/ethnic background.⁵¹ Taken together, these

Preferences related to ADHD and its treatment

studies provide insight into factors that may influence patient, parent, and clinician preferences for ADHD treatment, such as treatment characteristics, effectiveness for specific symptoms, side effects, and respondent demographics. The variety of available treatment approaches and factors that can influence treatment preference underscores the importance of customizing treatment decisions based on the needs and preferences of each individual patient, as no single treatment approach will be suitable for all patients. Additional research on treatment and patient characteristics that influence preference could provide useful guidance for clinicians involved in choosing among treatment options for individual patients.

One significant gap identified in the current literature review is the minimal available research on treatment preferences of adults with ADHD. Although ADHD is often believed to be a disorder of childhood, symptoms such as inattention and impulsivity often persist into adulthood.^{14–16,61} Furthermore, pharmacological and psychosocial treatments are being developed, tested, and implemented in adults with ADHD.^{62–68} However, the current literature search identified only one study examining preferences of adults with ADHD, and this study was conducted with a small sample.⁴⁹ Since there is a wide range of potentially effective treatment approaches for adult ADHD, research is needed to understand the treatment preferences of this population.

Another limitation of this literature is that most studies did not use carefully developed and validated instruments to assess preference. Some studies used invalidated global items,⁴⁶ while others did not clearly describe the method of preference assessment.^{47,48} Since the introduction of the Food and Drug Administration guidance on patient-reported outcomes, there has been a growing awareness of the importance of using carefully developed instruments that are validated for use in the target population.⁶⁹ We recommend that future studies of ADHD treatment incorporate more rigorously developed preference assessment methods, which can be clearly described in published articles.

Despite limitations of the currently available literature, findings of this review suggest that preference assessment could provide a useful indication of patients' experiences with various treatment options. Across the 15 studies in this review, patients, parents, clinicians, and teachers were able to provide quantifiable preferences among multiple treatment options, and research has begun to identify treatment- and respondentrelated factors that influence these preferences. As research on preferences related to ADHD grows, findings may be applied in clinical decision-making. Although current ADHD treatment guidelines acknowledge that parents and families may play

a role in choosing a treatment,^{70,71} no guidance based on preference research is provided. As preference data accumulate in ADHD studies, findings could be incorporated into the decision-making process as described in treatment guidelines. Furthermore, such guidelines could encourage clinicians to include patients and families in the decision-making process. Additional research findings may help clinicians know how to initiate and facilitate these discussions. Finally, decision aids, such as booklets or Web sites, may be developed to provide information that will assist patients and parents as they contribute to their own treatment decisions.72 Such decision aids have helped patients with other conditions develop their treatment preferences based on knowledge and information.73 It is likely that patients and families affected by ADHD may experience similar benefits. ADHD may be particularly appropriate for consideration of patient preferences in the use of decision aids because it is a condition with a range of potentially effective treatment options. ADHD can be addressed with behavioral treatment, stimulant medication, nonstimulant medication, and a combination of behavioral and pharmacological treatments. Treatment approaches that help educate patients and parents while considering their preferences may be more effective than treatment decisions based on efficacy alone.

Acknowledgments

The authors thank Aria Gray for production assistance. This study was funded by Eli Lilly and Company.

Disclosure

The authors report no conflicts of interest in this work.

References

- 1. Kon AA. The shared decision-making continuum. *JAMA*. 2010;304(8): 903–904.
- Murray E, Pollack L, White M, Lo B. Clinical decision-making: patients' preferences and experiences. *Patient Educ Couns*. 2007;65(2): 189–196.
- Keirns CC, Goold SD. Patient-centered care and preference-sensitive decision making. *JAMA*. 2009;302(16):1805–1806.
- Mandelblatt JS, Sheppard VB, Hurria A, et al. Breast cancer adjuvant chemotherapy decisions in older women: the role of patient preference and interactions with physicians. *J Clin Oncol*. 2010;28(19):3146–3153.
- Flood AB, Wennberg JE, Nease RF Jr, Fowler FJ Jr, Ding J, Hynes LM. The importance of patient preference in the decision to screen for prostate cancer. Prostate Patient Outcomes Research Team. *J Gen Intern Med.* 1996;11(6):342–349.
- Unic I, Verhoef LC, Stalmeier PF, van Daal WA. Prophylactic mastectomy or screening in women suspected to have the BRCA1/2 mutation: a prospective pilot study of women's treatment choices and medical and decision-analytic recommendations. *Med Decis Making*. 2000;20(3): 251–262.
- Meltzer EO, Andrews C, Journeay GE, et al. Comparison of patient preference for sensory attributes of fluticasone furoate or fluticasone propionate in adults with seasonal allergic rhinitis: a randomized, placebo-controlled, double-blind study. *Ann Allergy Asthma Immunol*. 2010;104(4):331–338.

- Kocsis JH, Leon AC, Markowitz JC, et al. Patient preference as a moderator of outcome for chronic forms of major depressive disorder treated with nefazodone, cognitive behavioral analysis system of psychotherapy, or their combination. *J Clin Psychiatry*. 2009;70(3):354–361.
- Dowson A, Bundy M, Salt R, Kilminster S. Patient preference for triptan formulations: a prospective study with zolmitriptan. *Headache*. 2007; 47(8):1144–1151.
- Stockl K, Ory C, Vanderplas A, et al. An evaluation of patient preference for an alternative insulin delivery system compared to standard vial and syringe. *Curr Med Res Opin*. 2007;23(1):133–146.
- Gold DT, Safi W, Trinh H. Patient preference and adherence: comparative US studies between two bisphosphonates, weekly risedronate and monthly ibandronate. *Curr Med Res Opin*. 2006;22(12):2383–2391.
- 12. American Psychiatric Association. *Diagnostic and Statistical Manual* of Mental Disorders. 4th ed. Washington, DC: American Psychiatric Press; 2000.
- Biederman J, Mick E, Faraone S, et al. A double-blind comparison of galantamine hydrogen bromide and placebo in adults with attentiondeficit/hyperactivity disorder: a pilot study. *J Clin Psychopharmacol*. 2006;26(2):163–166.
- Faraone SV, Biederman J. What is the prevalence of adult ADHD? Results of a population screen of 966 adults. *J Atten Disord*. 2005; 9(2):384–391.
- Kessler RC, Adler LA, Barkley R, et al. Patterns and predictors of attention-deficit/hyperactivity disorder persistence into adulthood: results from the national comorbidity survey replication. *Biol Psychiatry*. 2005;57(11):1442–1451.
- Mannuzza S, Klein RG, Moulton JL 3rd. Persistence of attention-deficit/ hyperactivity disorder into adulthood: what have we learned from the prospective follow-up studies? *J Atten Disord*. 2003;7(2):93–100.
- Bowling A, Ebrahim S. Measuring patients' preferences for treatment and perceptions of risk. *Qual Health Care*. 2001;10 Suppl 1:i2–i8.
- Brennan PF, Strombom I. Improving health care by understanding patient preferences: the role of computer technology. *J Am Med Inform Assoc.* 1998;5(3):257–262.
- Montgomery AA, Fahey T. How do patients' treatment preferences compare with those of clinicians? *Qual Health Care*. 2001;10 Suppl 1: i39–i43.
- Sidani S, Miranda J, Epstein DR, Bootzin RR, Cousins J, Moritz P. Relationships between personal beliefs and treatment acceptability, and preferences for behavioral treatments. *Behav Res Ther.* 2009;47(10): 823–829.
- Díez FI, Straube A, Zanchin G. Patient preference in migraine therapy. A randomized, open-label, crossover clinical trial of acute treatment of migraine with oral almotriptan and rizatriptan. *J Neurol.* 2007;254(2): 242–249.
- 22. Matza LS, Boye KS, Yurgin N, et al. Utilities and disutilities for type 2 diabetes treatment-related attributes. *Qual Life Res.* 2007;16(7): 1251–1265.
- Gudagnoli E, Ward P. Patient participation in decision making. Soc Sci Med. 1998(47):329–339.
- Brody DS, Miller SM, Lerman CE, Smith DG, Caputo GC. Patient perception of involvement in medical care: relationship to illness attitudes and outcomes. J Gen Intern Med. 1989;4(6):506–511.
- Eisenthal S, Emery R, Lazare A, Udin H. "Adherence" and the negotiated approach to patienthood. *Arch Gen Psychiatry*. 1979;36(4):393–398.
- Lazare A, Eisenthal S, Wasserman L. The customer approach to patienthood. Attending to patient requests in a walk-in clinic. *Arch Gen Psychiatry*. 1975;32(5):553–558.
- Raue PJ, Schulberg HC, Heo M, Klimstra S, Bruce ML. Patients' depression treatment preferences and initiation, adherence, and outcome: a randomized primary care study. *Psychiatr Serv.* 2009;60(3):337–343.
- Torrance GW. Preferences for health outcomes and cost-utility analysis. *Am J Manag Care*. 1997;3 Suppl:S8–S20.
- Brazier J, Ratcliffe J, Tsuchiya A, Salomon J. Measuring and Valuing Health Benefits for Economic Evaluation. New York, NY: Oxford University Press; 2007.

- Werz MA, Schoenberg MR, Meador KJ, et al. Subjective preference for lamotrigine or topiramate in healthy volunteers: relationship to cognitive and behavioral functioning. *Epilepsy Behav.* 2006;8(1): 181–191.
- Kendler D, Kung AW, Fuleihan Gel H, et al. Patients with osteoporosis prefer once weekly to once daily dosing with alendronate. *Maturitas*. 2004;48(3):243–251.
- Loder E, Brandes JL, Silberstein S, et al. Preference comparison of rizatriptan ODT 10-mg and sumatriptan 50-mg tablet in migraine. *Headache*. 2001;41(8):745–753.
- Berner MM, Kriston L, Sitta P, Härter M. Treatment of depressive symptoms and attitudes towards treatment options in a representative German general population sample. *Int J Psychiatry Clin Pract.* 2008; 12(1):5–10.
- Byrne CM, Solomon MJ, Young JM, Selby W, Harrison JD. Patient preferences between surgical and medical treatment in Crohn's disease. *Dis Colon Rectum*. 2007;50(5):586–597.
- Duarte JW, Bolge SC, Sen SS. An evaluation of patients' preferences for osteoporosis medications and their attributes: the PREFER-International study. *Clin Ther.* 2007;29(3):488–503.
- Peres MF, Silberstein S, Moreira F, et al. Patients' preference for migraine preventive therapy. *Headache*. 2007;47(4):540–545.
- Torrance GW. Utility approach to measuring health-related quality of life. J Chronic Dis. 1987;40(6):593–603.
- Torrance GW, Feeny D. Utilities and quality-adjusted life years. Int J Technol Assess Health Care. 1989;5(4):559–575.
- Bussing R, Schoenberg NE, Perwien AR. Knowledge and information about ADHD: evidence of cultural differences among African-American and white parents. *Soc Sci Med.* 1998;46(7):919–928.
- Bussing R, Gary FA, Mills TL, Garvan CW. Parental explanatory models of ADHD: gender and cultural variations. *Soc Psychiatry Psychiatr Epidemiol.* 2003;38(10):563–575.
- Shaw K, Wagner I, Eastwood H, Mitchell G. A qualitative study of Australian GPs' attitudes and practices in the diagnosis and management of attention-deficit/hyperactivity disorder (ADHD). *Fam Pract.* 2003;20(2):129–134.
- Chan E, Rappaport LA, Kemper KJ. Complementary and alternative therapies in childhood attention and hyperactivity problems. *J Dev Behav Pediatr.* 2003;24(1):4–8.
- Wood JC, Heiskell KD, Delay DM, Jongeling JA, Perry D. Teachers' preferences for interventions for ethnically diverse learners with attention-deficit hyperactivity disorder. *Adolescence*. 2009;44(174): 273–288.
- 44. Pisecco S, Huzinec C, Curtis D. The effect of child characteristics on teachers' acceptability of classroom-based behavioral strategies and psychostimulant medication for the treatment of ADHD. *J Clin Child Psychol.* 2001;30(3):413–421.
- Power TJ, Hess LE, Bennett DS. The acceptability of interventions for attention-deficit hyperactivity disorder among elementary and middle school teachers. J Dev Behav Pediatr. 1995;16(4):238–243.
- Efron D, Jarman F, Barker M. Methylphenidate versus dexamphetamine in children with attention deficit hyperactivity disorder: a double-blind, crossover trial. *Pediatrics*. 1997;100(6):E6.
- 47. Quintana H, Cherlin EA, Duesenberg DA, et al. Transition from methylphenidate or amphetamine to atomoxetine in children and adolescents with attention-deficit/hyperactivity disorder – a preliminary tolerability and efficacy study. *Clin Ther*. 2007;29(6):1168–1177.
- Pelham WE, Gnagy EM, Burrows-Maclean L, et al. Once-a-day Concerta methylphenidate versus three-times-daily methylphenidate in laboratory and natural settings. *Pediatrics*. 2001;107(6):E105.
- Fredericks EM, Kollins SH. Assessing methylphenidate preference in ADHD patients using a choice procedure. *Psychopharmacology (Berl)*. 2004;175(4):391–398.
- MacDonald Fredericks E, Kollins SH. A pilot study of methylphenidate preference assessment in children diagnosed with attention-deficit/ hyperactivity disorder. *J Child Adolesc Psychopharmacol*. 2005;15(5): 729–741.

- Dos Reis S, Zito JM, Safer DJ, Soeken KL, Mitchell JW Jr, Ellwood LC. Parental perceptions and satisfaction with stimulant medication for attention-deficit hyperactivity disorder. *J Dev Behav Pediatr*. 2003; 24(3):155–162.
- Glass CS, Wegar K. Teacher perceptions of the incidence and management of the attention deficit hyperactivity disorder. *Education*. 2000; 121(2):412–420.
- McLeod JD, Fettes DL, Jensen PS, Pescosolido BA, Martin JK. Public knowledge, beliefs, and treatment preferences concerning attentiondeficit hyperactivity disorder. *Psychiatr Serv.* 2007;58(5):626–631.
- Pham AV, Carlson JS, Kosciulek JF. Ethnic differences in parental beliefs of attention-deficit/hyperactivity disorder and treatment. *J Atten Disord*. 2010;13(6):584–591.
- Muhlbacher AC, Rudolph I, Lincke HJ, Nubling M. Preferences for treatment of attention-deficit/hyperactivity disorder (ADHD): a discrete choice experiment. *BMC Health Serv Res.* 2009;9:149.
- Matza LS, Secnik K, Rentz AM, et al. Assessment of health state utilities for attention-deficit/hyperactivity disorder in children using parent proxy report. *Qual Life Res.* 2005;14(3):735–747.
- Secnik K, Matza LS, Cottrell S, Edgell E, Tilden D, Mannix S. Health state utilities for childhood attention-deficit/hyperactivity disorder based on parent preferences in the United Kingdom. *Med Decis Making*. 2005; 25(1):56–70.
- Stockl KM, Hughes TE, Jarrar MA, Secnik K, Perwien AR. Physician perceptions of the use of medications for attention deficit hyperactivity disorder. *J Manag Care Pharm.* 2003;9(5):416–423.
- Sciberras E, Iyer S, Efron D, Green J. Information needs of parents of children with attention-deficit/hyperactivity disorder. *Clin Pediatr* (*Phila*). 2010;49(2):150–157.
- Bhatara VS, Vogt HB, Patrick S, Doniparthi L, Ellis R. Acceptability of a Web-based attention-deficit/hyperactivity disorder scale (T-SKAMP) by teachers: a pilot study. *J Am Board Fam Med.* 2006;19(2):195–200.
- Biederman J, Monuteaux MC, Mick E, et al. Young adult outcome of attention deficit hyperactivity disorder: a controlled 10-year follow-up study. *Psychol Med.* 2006;36(2):167–179.

- Davidson MA. ADHD in adults: a review of the literature. J Atten Disord. 2008;11(6):628–641.
- Kolar D, Keller A, Golfinopoulos M, Cumyn L, Syer C, Hechtman L. Treatment of adults with attention-deficit/hyperactivity disorder. *Neuropsychiatr Dis Treat*. 2008;4(2):389–403.
- Safren SA, Sprich S, Chulvick S, Otto MW. Psychosocial treatments for adults with attention-deficit/hyperactivity disorder. *Psychiatr Clin North Am.* 2004;27(2):349–360.
- 65. Simpson D, Plosker GL. Atomoxetine: a review of its use in adults with attention deficit hyperactivity disorder. *Drugs*. 2004;64(2):205–222.
- Spencer T, Biederman J, Wilens T. Nonstimulant treatment of adult attention-deficit/hyperactivity disorder. *Psychiatr Clin North Am.* 2004; 27(2):373–383.
- Spencer T, Biederman J, Wilens T. Stimulant treatment of adult attention-deficit/hyperactivity disorder. *Psychiatr Clin North Am.* 2004; 27(2):361–372.
- Wilens TE. Pharmacotherapy of ADHD in adults. CNS Spectr. 2008; 13(5 Suppl 8):11–13.
- Food and Drug Administration. Guidance for Industry-Patient-Reported Outcome Measures: Use in Medical Product Development to Support Labeling Claims. Silver Spring, MD: FDA; 2009.
- Pliszka S; AACAP Work Group on Quality Issues. Practice parameter for the assessment and treatment of children and adolescents with attention-deficit/hyperactivity disorder. J Am Acad Child Adolesc Psychiatry. 2007;46(7):894–921.
- National Institute for Health and Clinical Excellence. Attention deficit hyperactivity disorder: diagnosis and management of ADHD in children, young people and adults. *NICE Clinical Guideline*. 2008;72:1–59.
- Edwards A, Elwyn G. The potential benefits of decision aids in clinical medicine. JAMA. 1999;282(8):779–780.
- O'Connor AM, Rostom A, Fiset V, et al. Decision aids for patients facing health treatment or screening decisions: systematic review. *BMJ*. 1999;319(7212):731–734.

Patient Preference and Adherence

Dovepress

Publish your work in this journal

Patient Preference and Adherence is an international, peer-reviewed, open access journal focusing on the growing importance of patient preference and adherence throughout the therapeutic continuum. Patient satisfaction, acceptability, quality of life, compliance, persistence and their role in developing new therapeutic modalities and compounds to optimize clinical outcomes for existing disease states are major areas of interest. This journal has been accepted for indexing on PubMed Central. The manuscript management system is completely online and includes a very quick and fair peer-review system. Visit http://www.dovepress.com/ testimonials.php to read real quotes from published authors.

Submit your manuscript here: http://www.dovepress.com/patient-preference-and-adherence-journal